Author's response to reviews

Title: Development and preliminary validation of a Greek-language outpatient satisfaction questionnaire with principal components and multi-trait analyses.

Authors:

Vassilis H Aletras (valetras@uom.gr)  
Efthimis A Papadopoulos (efthimiospapado@hotmail.com)  
Dimitris A Niakas (niakas@eap.gr)

Version: 3 Date: 8 May 2006

Author's response to reviews:

Dear Editors,

We submit the final version of the accepted manuscript "Development and preliminary validation of a Greek-language outpatient satisfaction questionnaire with principal components and multi-trait analyses". A point-by-point description of the changes made in light of the referees' comments is provided. All comments relate to the language and presentation of the text.

Jose Labarere's comments:

A. Minor essential revisions:
(1) We have deleted the sentence suggested from the specific section and moved it to the "Discussion" section (p. 18, 1st paragraph).
(2) The figure missing has been included in Table 3 (p. 30).

B. Discretionary revisions:
(3) We believe our choice of the specific intraclass correlation is adequately explain and chose not to make any further comment on the specific issue. We note however that this refers to a change that the reviewer himself classifies as one that "the author can choose to ignore".
(4) We agree with the reviewer and made Table 5 easier to read (p. 33-34).

Andrew Garratt's comments:
We read Dr. Garratt's grammar suggestions in the copy of the manuscript sent to us and made all necessary changes in the text. In addition, we also corrected a few remaining errors in the format and grammar ourselves. We once more thank the reviewers for this final contribution.

Yours faithfully,

Vassilis Aletras